Author's response to reviews

Title: MDM2 SNP309 promoter polymorphism and p53 mutations in urinary bladder carcinoma stage T1

Authors:

Hans Olsson (hansolsson33@telia.com)
Per Hultman (Per.Hultman@liu.se)
Johan Rosell (Johan.Rosell@liu.se)
Peter Söderkvist (peter.soderkvist@liu.se)
Staffan Jahnson (Staffan.Jahnson@liu.se)

Version: 2 Date: 29 October 2012

Author's response to reviews:

Dear Editor.

As requested we have now included that this research have ethical approval, and we have stated this in the method part of the manuscript.

Kind regards

Yours Sincerely

Dr Hans Olsson